Author's response to reviews

Title: Continuum of care for frail elderly people: Design of a randomized controlled study of a multi-professional and multidimensional intervention targeting frail elderly people.

Authors:

Katarina Wilhelmson (katarina.wilhelmson@socmed.gu.se)
Anna Duner (anna.duner@socwork.gu.se)
Kajsa Eklund (kajsa.eklund@neuro.gu.se)
Gunilla Gosman-Hedström (gunilla.gosman-hedstrom@neuro.gu.se)
Staffan Blomberg (staffan.blomberg@soch.lu.se)
Henna Hasson (henna.hasson@fek.lu.se)
Helena Gustafsson (helena.a.gustafsson@vgregion.se)
Sten Landahl (sten.landahl@vardalinstitutet.net)
Synneve Dahlin-Ivanoff (synneve.dahlin-ivanoff@neuro.gu.se)

Version: 3 Date: 4 May 2011

Author's response to reviews: see over
The Editor,

*BMC Geriatrics*

Concerning:
MS: 5279877285088799

Study protocol
Continuum of care for frail elderly people: Design of a randomized controlled study of a multi-professional and multidimensional intervention targeting frail elderly people.
Katarina Wilhelmson, Anna Duner, Kajsa Eklund, Gunilla Gosman-Hedström, Staffan Blomberg, Henna Hasson, Helena Gustafsson, Sten Landahl and Synneve Dahlin-Ivanoff

BMC Geriatrics

Dear Editor,

The manuscript (study protocol) entitled “Continuum of care for frail elderly people: Design of a randomized controlled study of a multi-professional and multidimensional intervention targeting frail elderly people” has been revised and resubmitted.

Revisions:
1) The author list is now with First name Initial Last name.
2) All tracking are removed.
3) All vertical lines in the tables are removed
4) The title is removed from the figure file

Yours sincerely

Katarina Wilhelmson
Sahlgrenska Academy at University of Gothenburg
Department of Public Health and Community Medicine/ Social Medicine,
P O Box 453
S-405 30 Göteborg
Sweden
Telephone: +46-31 786 68 73
Telex: +46-31 16 28 47
E-mail: katarina.wilhelmson@socmed.gu.se